Author’s response to reviews

Title: Value of information methods to design a clinical trial in a small population to optimise a health economic utility function

Authors:

Michael Pearce (m.a.l.pearce@warwick.ac.uk)
Siew Wan Hee (s.w.hee@warwick.ac.uk)
Jason Madan (j.j.madan@warwick.ac.uk)
Martin Posch (Martin.Posch@meduniwien.ac.at)
Simon Day (simon.day@ctct-ltd.co.uk)
Frank Miller (mail@frank-miller.eu)
Sarah Zohar (sarah.zohar@inserm.fr)
Nigel Stallard (N.Stallard@warwick.ac.uk)

Version: 2 Date: 15 Dec 2017

Author’s response to reviews:

Additional points regarding clarification of author contributions and details of files for the supplementary material have been addressed